Author’s response to reviews

Title: Efficacy of sorafenib on metastatic renal cell carcinoma in Asian patients: Results from a multicenter study

Authors:

Hailiang Zhang (zhanghl918918@yahoo.com.cn)
Baijun Dong (dongbaijun@hotmail.com)
Jiade J Lu (sintelligence@gmail.com)
Xudong Yao (xudongyao67@shca.org.cn)
Shilin Zhang (zhangss1969@shca.org.cn)
Bo Dai (bodai1978@shca.org.cn)
Yijun Shen (yijunshen@shca.org.cn)
Yao Zhu (zhu_yao@shca.org.cn)
Dingwei Ye (dwye@shca.org.cn)
Yiran Huang (hyrrenji2@yahoo.com.cn)

Version: 6 Date: 14 July 2009

Author’s response to reviews:

Dear Editors,

Thank you for accepting our manuscript. We have made editing changes according to your direction. All figures are uploaded separately and descriptions relocated to the end of the manuscript after the reference section. All tables are also inserted after reference as directed. The minor revisions were also completed according to your direction.

Please let us know if further editing is needed.

Best,

Professors Ye and Huang.